Author's response to reviews

Title: Use of an Electromagnetic Colonoscope to Assess Maneuvers Associated with Cecal Intubation

Authors:

Russell I. Heigh (heigh.russell@mayo.edu)
John K. Dibaise (dibaise.john@mayo.edu)
James A. Prechel (prechel.james@mayo.edu)
Billie J. Horn (horn.bilie@mayo.edu)
Sarah San Miguel (sanmiguel.sarah@mayo.edu)
Evelyn G. Heigh (evelyn.heigh@asu.edu)
Jonathan A. Leighton (leighton.jonathan@mayo.edu)
Cynthia G. Edgelow (edgelow.cynthia@mayo.edu)
David E. Fleischer (fleischer.david@mayo.edu)

Version: 3 Date: 8 March 2009

Author's response to reviews:

Dear Dr. Zauner and Ms. Awdry,

Thank you for your thoughtful consideration of our manuscript. The attached revision includes all 3 modifications specified by the reviewers. It is our hope that with the inclusion of these helpful suggestions our manuscript is now ready for publication.

The only changes made on this latest revision are those requested by Referee 1 in your latest communication to us:

1. In the abstract the actual detailed results about position of the instrument are now included in addition to the "most often" and "less". In the abstract these results were brought into the abstract from where they were previously in Table 2. Table 2 remains intact.

2. In the abstract the actual detailed results about success rates are now included in addition to the "most often" and "less". In the abstract these results were brought into the abstract from where they were previously in Table 2. Table 2 remains intact.

3. In the results section a detailed sentence has been added to describe the sedation plan and outcome in all of the patients.

Thank you very much for helping us develop the best manuscript possible. We look forward to seeing our manuscript as a published work on BioMed Central.

Sincerely,

Russell Heigh, MD, FACP, FACG, AGAF, FASGE
Associate Professor Of Medicine
Mayo Clinic
Asociate Dean, Mayo School Continuous Professional Development
13400 E. Shea Blvd.
Scottsdale, Arizona 85259
heigh.russell@mayo.edu
480-301-6990 phone
480-391-9123 fax
(Please make sure that I am the corresponding author - thank you)